Author's response to reviews

Title: The usefulness of school-based syndromic surveillance for detecting malaria epidemics: experiences from a pilot project in Ethiopia

Authors:

Ruth A Ashton (rashton@tulane.edu)
Takele Kefyalew (t.kefyalew@malariaconsortium.org)
Esey Batisso (e.batisso@malariaconsortium.org)
Tessema Awano (t.awano@malariaconsortium.org)
Zelalem Kebede (zelalemkebede@yahoo.com)
Gezahegn Tesfaye (g.tesfaye@yahoo.com)
Tamiru Mesele (tamex23@yahoo.com)
Sheleme Chibsa (schibsa@usaid.gov)
Richard Reithinger (r.reithinger@yahoo.co.uk)
Simon J Brooker (simon.brooker@gatesfoundation.org)

Version: 4 Date: 18 December 2015

Author's response to reviews: see over
Response to BMC Public Health editorial revisions on MS 1677620913166731: “The usefulness of school-based syndromic surveillance for detecting malaria epidemics: experiences from a pilot project in Ethiopia”

Dear Editor,

I can confirm that I generated the maps in Figure 2 of the above named manuscript myself using Geographic Information System (GIS) software. Therefore there are no additional permissions required to publish this figure.

Kind regards,

Ruth Ashton, PhD

Research Assistant Professor, Tulane University (Current)
Epidemiologist, Malaria Consortium, London (Previous)